Massachusetts Institute of Technology Handout 28 6.915: Network and Computer Security November 7, 1995 Professor Ronald L. Rivest

Readings for Lecture 17

Attached are the following papers:

ffl PayWord and MicroMint: Two simple micropayemnt schemes by Ron Rivest and Adi Shamir [2].

ffl The Millicent Protocols for Electronic Commerce by Mark Manasse (you can get to this paper by following reference [1].

References

[1] Mark S. Manasse. Millicent (electronic microcommerce), 1995.

http://www.research.digital.com/SRC/personal/Mark Manasse/uncommon/ucom.html.

[2] Ronald L. Rivest and Adi Shamir. Payword and micromint: Two simple micropayment schemes. (Submitted to Eurocrypt ’96.), 1995.